Author’s response to reviews

Title: EFFECT OF PEEP AND I:E RATIO ON CEREBRAL OXYGENATION IN ARDS: AN EXPERIMENTAL STUDY IN ANESTHETIZED RABBIT

Authors:

Federica Lovisari (federica.lovisari@gmail.com)
Gergely Fodor (Gergely.Fodor@hcuge.ch)
Ferenc Peták (petak@dmi.szote.u-szeged.hu)
Walid Habre (Walid.Habre@hcuge.ch)
Sam Bayat (sbayat@chu-grenoble.fr)

Version: 1 Date: 09 Oct 2018

Author’s response to reviews:

Dear Matthew Hickey,

As per your request, I have added the authors' email addresses in the title page. I also included the source of animals used in the study. Please don't hesitate letting me know if any further information is needed.

Best regards,

Sam Bayat

Sam Bayat, MD, PhD
Professor, Dept. of Clinical Physiology, Sleep and Exercise
CS 10217, 38043 Grenoble Cedex 9, France

Director, Synchrotron Radiation for Medicine Laboratory - EA 7442
University of Grenoble Alpes – ID17 Biomedical Beamline, European Synchrotron Radiation Facility (ESRF)
CS 40220, 38043 Grenoble Cedex 9, France

Office : +33 (0)4 76 76 77 26
Mobile : +33 (0)6 14 97 17 09
sbayat@chu-grenoble.fr